Author's response to reviews

Title: Neck muscle cross-sectional area, brain volume and cognition in healthy older men; a cohort study

Authors:

Alixe HM Kilgour (a.kilgour@ed.ac.uk)
Karen J Ferguson (karen.ferguson@ed.ac.uk)
Calum D Gray (calum.gray@ed.ac.uk)
Ian J Deary (iand@staffmail.ed.ac.uk)
Joanna M Wardlaw (joanna.wardlaw@ed.ac.uk)
Alasdair MJ MacLullich (a.macullich@ed.ac.uk)
John M Starr (jstarr@staffmail.ed.ac.uk)

Version: 4 Date: 6 November 2012

Author's response to reviews:

We were asked by the editorial team to include a statement regarding informed consent, state that the research was carried out in compliance with the Declaration of Helsinki and state the name of the ethics committee which gave approval for the study. We have now concluded this information in the methods section. We have also now included a paragraph stating the authors' contributions.